		EO-1 Weekly Report
	Earth Observing – 1

	August 10, 2016 – August 16, 2016

	DOY
	223
	–
	229

	Mission Day
	5791
	–
	5797


EO-1 SPACECRAFT

Instruments
· Scheduled 118 science Data Collection Events (DCEs) 
· Loss of [1] DCE, see the operational discrepancies section below for more details
· Performed ALI and HYP out gassing on DOY 228/16:55z-229/07:55z
· Performed Instrument Calibration
· Conducted HYP solar cal on DOY 228/01:55z
· Conducted SA Characterization on DOY 229/02:21z

EO-1 MISSION OPERATIONS CENTER
Real-Time
· Maintaining nominal systems state of health 
· Monitoring the battery data
· Monitoring the clock data
· Adjusting the clock as necessary 
· Successfully tested the manual 1Hz Delta Adjust of 2mS
· Looking into ways of automating this process 
· Adjusted the 1Hz slope for the 1Hz Auto Adjust
· Took a test support with USAK04 on DOY 228/19:59z 
· Initially due to AK3 shadowing the support, there were sequence breaks in the data. The controllers at USN stopped the shadowing of AK3, and the sequence breaks and the multiple connections to our command port stopped.  
· S-band and X-band data looked fine
· The tracking data file was successfully verified on the backup FD machine. 
· Dumped the EEPROM RTS’s and compared the content with RTS’s dumped from the RAM
· There are differences; the differences do not matter though because during passivation new RTS’s will be uploaded anyway. 
· Working on creating a decommissioning sequence
· Created (4) rough drafts of the sequences 
· Created (2) passivation related RTS’s
· Demonstrated SCP’s ability of creating RTS’s 
· Researching the ability to write to EEPROM for passivation
· Either upload new RTS’s to EEPROM or patch the boot code to contain a halt instruction 
· Created a procedure to perform 1HZ UTCF DOT adjust
· Verifying the procedures on all adjustments  
· Adding a UTCF adjustment section 
· AS1 and USAK04 certification supports are being scheduled for the coming weeks
· Working on upgrading the FD windows XP machines to windows 7
· See FD section of this report for more details 
· [bookmark: _GoBack]Testing Overdraw after update in downtime of FD systems
· Encountered problem with AT command being 1 hour off
· Encountered problem with the system directory change on update, having to adjust all scripts that reference rstol

Mission Planning
· Scheduled a USAK04 test support on DOY 228/19:59z
· Created a ACNOOP command RTS 
· To test for an UPLOAD to the EEPORM and RAM
· Created RTS’s with EEPROM as destination 
· Verifying and selecting AS1/AS3 S- and S/X- band supports
· Adding in AS 1/3 ground stations to ASPEN
· Working with the USAK04 personnel to bring test passes with that ground station

Flight Dynamics 
· Added USAK04 as a tracking ground station on the backup FD machine
· Received and tested USAK04 TDF data
· Testing AS1 and AS3 tracking data
· AS1 tracking data certification will begin during S- and X-band certification
· AS3 tracking data certification has completed successfully

Central File Hub
· Phase one tests are ready for outside entity delivery.
· USAK03, AS1, AS3 all deliver to file hubs and its working fine
· SGS and WGS tracking data are waiting for a break in Wallops activities to begin testing.
· Phase 2 FD scripts have no changes from Phase 1 scripts
· All non-FD scripts are implemented Phase 2 
· Beginning Phase 3 work on deliveries for the central file hub
· Central file hub delivery structure set
· Passwordless ssh set up from FD to eo1out and eo1in to and from eo1out
· Script to put FD products onto file hubs is complete and ready for use
· Script to sort FD products into destination directories is complete and ready for use
· Base delivery script completed
· Designing delivery code architecture
· Reviewing current deliveries for products, emails, and destinations
· Delivery will be completed via FD computers as is the current standard

System Administration
· Resolved the issue with the habanero computer showing up on the KACE August block list. The habanero computer is now reporting its inventory to KACE.
· Emailed James Cameron to expedite firewall rule request #011-0012, allowing DPS04 and DPS05 outbound port 22 access to the new EDC host edcsns19.
· Firewall rule #011-0012 was approved and implemented.
· Rebooted the clock computer.
· Completed the SATERN training course Protecting Windows 7 against Malware and Vulnerabilities.
· Renewed the expiration dates to 2017 for the following EO-1 firewall rules in the MSGRS database: #011-0012, #011-0014, #011-0015, #011-0138, and #012-0013.
· Attended the Flight Project Directorate weekly SA meeting.
· Opened MOCR #430 to apply August Windows updates to the MOC Windows computers.  These updates include 2 expedited security patches. 
· Created the weekly backup tape containing backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, and regional logs server.
· Created weekly backups of Statler2, Waldorf2, eo1itps1, and eo1itps2.
· Created the monthly backup of the MOC CNE computer. 

GROUND AND SPACE NETWORK
Station Downtimes 	
· None to report at this time

[bookmark: _MON_1387176949][bookmark: _MON_1388382035][bookmark: _MON_1388382145][bookmark: _MON_1388384576][bookmark: _MON_1388385083][bookmark: _MON_1388993005][bookmark: _MON_1389598124][bookmark: _MON_1390809671][bookmark: _MON_1390809790][bookmark: _MON_1391499119][bookmark: _MON_1391501949][bookmark: _MON_1392188921][bookmark: _MON_1392615265][bookmark: _MON_1393314853][bookmark: _MON_1393834790][bookmark: _MON_1394437595][bookmark: _MON_1394437598][bookmark: _MON_1394437644][bookmark: _MON_1395039355][bookmark: _MON_1395831044][bookmark: _MON_1396332066][bookmark: _MON_1396332146][bookmark: _MON_1396332643][bookmark: _MON_1396332666][bookmark: _MON_1396332684][bookmark: _MON_1396332702][bookmark: _MON_1396862017][bookmark: _MON_1397634356][bookmark: _MON_1397634698]Operational Discrepancies:
· Downlink: 2016:225:23:55:10 2016:226:00:04:30 TR2
· Problem:  Appears to be late start of X-band collection at station. First recorded data was at 20162252359, part way thru dump of second scene. Length of contact did not allow complete second dump to be downlinked to station.
· Result:   Loss of one [1] scene





UPCOMING EVENTS
· Testing and validation of FD systems on Windows 7, ASIST systems on RedHat upgrade
· Testing and certification with AS1 and USAK04
· Continuation of Phase 1/2 FD, waiting on WALLOPS support 
· Moving on to Phase 3 for central file hub, deliveries
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Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

10-Aug-16 223 5 2 0 0 7

11-Aug-16 224 6 3 0 0 9

12-Aug-16 225 4 3 0 0 7

13-Aug-16 226 6 3 0 0 9

14-Aug-16 227 5 4 0 0 9

15-Aug-16 228 4 5 0 0 9

16-Aug-16 229 2 7 0 0 9

32 27 0 0 59 Weekly Totals
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		10-Aug-16		223		5		2		0		0		7

		11-Aug-16		224		6		3		0		0		9

		12-Aug-16		225		4		3		0		0		7

		13-Aug-16		226		6		3		0		0		9

		14-Aug-16		227		5		4		0		0		9

		15-Aug-16		228		4		5		0		0		9

		16-Aug-16		229		2		7		0		0		9

		Weekly Totals				32		27		0		0		59
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